Author’s response to reviews

Title: Acromioclavicular joint dislocation treated with Bosworth Screw and additional K-wire fixation: Results after in mean 7.8 years - Still an adequate procedure?

Authors:

Thomas Tiefenboeck (thomas.tiefenboeck@meduniwien.ac.at)
Domenik Popp (domenik.popp@hotmail.com)
Sandra Boesmueller (sandra.boesmueller@auva.at)
Stephan Payr (stephan.payr@meduniwien.ac.at)
Julian Joestl (julian.joestl@meduniwien.ac.at)
Micha Komjati (micha.komjati@gmail.com)
Harald Binder (harald.binder@meduniwien.ac.at)
Mark Schurz (mark.schurz@meduniwien.ac.at)
Roman Ostermann (roman.ostermann@meduniwien.ac.at)

Version: 4 Date: 18 Jul 2017

Author’s response to reviews:

BMC Musculoskeletal Disorders

Editor in Chief: Umile Giuseppe Longo, M.D.

BMSD-D-16-00842R2

Acromioclavicular joint dislocation treated with Bosworth Screw and additional K-wire fixation: Results after in mean 7.8 years - Still an adequate procedure?

Thomas Tiefenboeck, M.D.; Domenik Popp, M.D.; Sandra Boesmueller, M.D.; Stephan Payr, M.D.; Julian Joestl, M.D.; Micha Komjati, MBBS; Harald Binder, M.D.; Mark Schurz, M.D.; Roman C Ostermann, M.D.
Dear Umile Giuseppe Longo, M.D.

Thank you very much for your letter of 17th July 2017. We have carefully followed the verbatim comments of the reviewers and are enclosing a revised manuscript. All performed changes are marked in yellow for better identification. Responses to reviewers are listed below in detail.

1. The authors' contribution statement you have submitted is not specific and must be revised. Please do not use the phrase "and/or" in this section, but please definitively state what each author contributed to this project.

Answer: Changes were made as suggested

2. Please delete supplementary files, "Response_to_Reviewers_BMC_Bosworth_REV3.docx" and "Letter_to_the_Editor_BMC_Bosworth_REV3.docx". They are no longer necessary at this stage of production.

Answer: Files were deleted

3. Upon uploading your revisions, please remove any tracked changes or highlighting and include only a single clean copy of the manuscript.

Answer: A single clean copy was uploaded